MCC Operations Status Report:  24-Jan-2008

In this mornings 8am meeting:

Safety: 
They are moving air conditioners on the MCC lower parking lot.

Software Day Status:

1. Phase came back up yesterday at 30deg off but not sure what caused this. This problem has occurred before and not sure if the cause is hardware, software or ioc save/restore issue.

Software

Bunch Length: Mike Zelanzy

1. Background image work yesterday ran into a problem, because the images from the laser did not have the correct timestamp. Mike implemented a workaround but has had an opportunity to test but will try to coordinate testing time this afternoon. Final solution is EVR software upgrade that Stephanie started to implement yesterday. Do not know status of this upgrade yet.

Computer Systems: Jingchen

1. Artemis down.

2. Pepii luminosity display down

3. The Firefox change done yesterday to distinguish the difference between the Firefox that runs on lcls-srv01 and the other one that runs on lcls-prod02, doesn’t work.

4. The Linux boxes in MCC have a problem with the key mapping, which is relatively new. The right arrow key deletes characters. This mapping occurs on a console window and in matlab.

Feedback: Diane Fairley

Implementation (requirement change) of feedback may need to be altered. Joe stopped/started and the feedback picked up the value prior to stopping feedback. This is the feature that Joe may want to change or to decide if a procedure will instead need to be changed.
Laser: 

1. Pulse with shorter as a result of work done yesterday by laser team.

LLRF: Ron Akre, Dayle Kotturi and Vojtech

1. X-band feedback and TCAV3 feedback work done yesterday by Dayle works well and physicists are quite happy. The feedback was considerable faster.

2. The RF System Alignment software (matlab), written by Vojtech needs the GUI updated with a similar change that Dayle added yesterday. Joe requested that the matlab program be made production worthy including any minimal changes and/or testing necessary, because the physicists would like to switch over to this new software.

SLC Database: Ken Underwood 
1. DbInstall scheduled today at 10-11am. All slc-aware iocs will need to be rebooted after the install is complete. This include, bpm’s, magnets and toroids.

SCORE: Debbie Rogind

1. Tried to "open" the GOLD saved injector configuration, but did not get data. Solution: Problem was a local physics account java file problem. This local file specifies that EPICS server and repeater ports, and this was changed to the development ports. This was done on Jan 22, and once before on Jan 8th. Debbie updated the file and changed the permissions to no write.

SLC Processes: Nancy Spencer

1. Fix SLCCAS. When SLCCAS was backed out to an August 2007 version, the image tied to this release was not backed out. The old image was retrieved and installed by the VMS team. 

Wire Scanners: Arturo Alarcon and Doug McCormick.

1. WS01 is still not working properly. 

2. The LI28 wire scanner’s need to be timed-in. After the meeting Doug went to li28 to time-in wire scanner

3. Paul E. would like a wire scanner fest to discuss a plan for completing the wire scanner commissioning.

